\documentclass [12pt] {article}

\usepackage{graphicx, amsmath, amsthm, amssymb, hyperref}

\theoremstyle{definition}
\newtheorem{definition}{Definition}[section]

\topmargin 0pt
\advance \topmargin by -\headheight
\advance \topmargin by -\headsep
\textheight 8.9in
\oddsidemargin 0pt
\evensidemargin \oddsidemargin
\marginparwidth 0.5in
\textwidth 6.5in

\newenvironment{packed_itemize}{
\begin{itemize}
  \setlength{\itemsep}{1pt}
  \setlength{\parskip}{0pt}
  \setlength{\parsep}{0pt}
}{\end{itemize}}


\newenvironment{packed_enum}{
\begin{enumerate}
  \setlength{\itemsep}{1pt}
  \setlength{\parskip}{0pt}
  \setlength{\parsep}{0pt}
}{\end{enumerate}}

\begin{document}
\begin{center}

\Huge{3D Printing HW5}
\end{center}


\normalsize


\noindent Each \emph{group} must:

\begin{enumerate}
\item $[2]$ bring the most recent 3D printed object created for your project and be able to:
\begin{enumerate}
\item $[2]$ explain to the class what is successful about the print, and
\item $[2]$ explain to the class what still has to be done.
\end{enumerate}
\item turn in a paper with:
\begin{enumerate}
\item $[2]$ a screen shot of the 3D file you have been working with, and
\item $[2]$ a draft of the material that you intend to put on the poster.
\end{enumerate}
\end{enumerate}


\end{document}